Author's response to reviews

Title: Community participation to design rural primary healthcare services

Authors:

Jane Farmer (j.farmer@latrobe.edu.au)
Amy Nimegeer (a.nimegeer@sphsu.mrc.ac.uk)

Version: 5 Date: 7 March 2014

Author's response to reviews: see over
How reviewers comments were addressed

We have reworded the acknowledgements and the sentence containing the information about the 39 interviews as suggested by the reviewer.